
-----------------------------------------------------------------------------
          Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
       Tue Jul 21 16:35:24 GMT 2015
Year: 15  Doy: 202
Observer: stueben
WEATHER COMMENT: stueben: Tue Jul 21 16:35:30 GMT 2015
Clear sky, nasty vog layer up to at least 10,000ft, light eastern breeze,
temp 42F. Boardwalk slippery with frost.
___end___
Tue Jul 21 17:09:19 GMT 2015 KCOR Start Synoptic Patrol
Tue Jul 21 17:23:34 GMT 2015 COMP Start Patrol
**PSPT PROBLEM COMMENT BY stueben** : Tue Jul 21 17:24:39 GMT 2015
The new cables arrived at my house on Friday and I have them with me today. 
Taking several tripes to bring them up from the parking lot and will install 
them today.
___end___
Tue Jul 21 18:11:46 GMT 2015 KCOR End Patrol
Tue Jul 21 18:11:48 GMT 2015 KCOR Start Calibration script: c:\kcor\mlso-calibration22deg-20150323.ini
Tue Jul 21 18:29:11 GMT 2015 KCOR End Calibration Script
Tue Jul 21 18:29:28 GMT 2015 KCOR Start Synoptic Patrol
Tue Jul 21 18:29:29 GMT 2015 KCOR Start Synoptic Patrol
Tue Jul 21 19:12:19 GMT 2015 KCOR End Patrol
Tue Jul 21 19:12:46 GMT 2015 COMP End Patrol
WEATHER COMMENT: stueben: Tue Jul 21 23:16:47 GMT 2015
In dense clouds, instruments idled.
___end___
KCOR COMMENT BY stueben: Wed Jul 22 00:15:03 GMT 2015
Taped black cloth around the cameras and optics in the aft section as requested
by Dennis to block out stray light. Also found a gap in the sheet metal on the
around the connector plate where light might enter and covered it with aluminum
tape. Will see if it makes any difference when I am up on Thursday.
___end___
**PSPT PROBLEM COMMENT BY stueben** : Wed Jul 22 00:37:41 GMT 2015
Replaced the old cables with the new cables, but have not tried to run pspt.
More on Thursday.
___end___


